Dive 4238 Transcript, Port Observer - Marv Lilley

Sept. 3, 2006

Pilot: Bruce Strickrott
Starboard Observer: Matt Lilley
GMT

1617

We are on the bottom getting neutral.
1618

Still at landing site x=5663, y=8405, headed for Godzilla.

1620
X=5684, y=8384, Just passed a double railroad wheel. Probably a Rick mooring anchor.

1622
We have found a sulfide structure but don’t see any old marker.

1625
x=5722, y=8350, We have found marker 1P so this structure is Three Sisters.

1631
x=5734, y=8334, still looking for Godzilla.

1637
x=5744, y=8300, at another structure but seems too small to be Godzilla.

1640
x=5762, y=8308, hdg 178, we are at Boardwalk looking around.

1648
x=5764, y=8299, setting up to sample Boardwalk.

1653
Temp prior to excavation was 346. After excavation the temp was 356. An incidental sulfide sample is in the front of the basket.
1702
We are water sampling at Boardwalk. X=5764, y=8299, z=2139, alt bogus, hdg 248. The yellow major pair is first followed by the yellow and blue gas tights. We also took a small sulfide sample.

1707
Finished water sampling and deploy triangle Marker Y near the sampled vent on Boardwalk.

1735
x=5789, y=8317, hdg 031, z=2141, alt 20, setting up to sample Godzilla. We take the red major pair, red gas tight, red/black gastight, do the toaster and collect a small sulfide sample. We also leave triangle Marker T by the sample site.
1758
x=5790, y=8311, leaving Godzilla area to hunt for Fairy Castle.

1811
Driving sort of aimlessly trying to find Fairy Castle. Our nav is acting up and top lab is not doing much better.

1813
We are trying to use top lab LBL to find Fairy Castle. 

1815
x=5764, y=8114 (These are Doppler fixes and are likely bogus. The terrain here is a mix of broken sheet flows with some areas of intact lobates.

1845
x=5696, y=8112, we have finally found Fairy Castle.

1857
We are set up to sample Fairy Castle, x=5697, y=8122, z=2148, alt=15, hdg=004. The high T probe gets 350.8 deg. We collect the blue major pair, the black and green gas tights and a small piece of sulfide.

1910
Finished sampling at Fairy Castle and head south to look for Huaiyang’s little baskets.

1922
x=5645, y=8026, z=2166, hdg 208, driving parallel to a basalt wall out the port window.

1928
x=5535, y=7892, z=2183, hdg 220, hackly sheet flow 

1929
Extinct sulfide, x=5530, y=7862, z=2181, dead tube worms around it.

1930
Small chimneys, x=5523, y=7856, diffuse flow. Significant chunk of old sulfide.

1932
Nav computer frozen up. Stop and reboot.

1938
x=5473, y=7841, these fixes are the result of resetting the Doppler to top lab LNL again. We are in an area of extensive tube worms and the top lab nav has us in the right area for the Chinese pagodas.

1948
x=5506, y=7805, active flange with black smoke overflowing. Still looking for baskets. This should be the right area.

2002
Give up on finding baskets and head south for our next target which is Lions Head.

2004
x=5481, y=7761, rubble ridge out my window. Z=2186.

2005
Back in tube worms. X=5465, y=7733. 

2006
We find stuff and think it is the baskets but it turns out to be an experiment left by Kim Juniper. We are in Clam Bed. Active sulfide around. X=5456, y=7711, z=2185 . We collect Kim’s experiment.
2014
We head south in an attempt to reach Raven.

2018
x=5461, y=7701, z=2185 driving south.
2025
x=5356, y=7546, z=2192 driving south along a basalt ridge, I can’t see bottom except the rubble right below the sub.

2026
x=5341, y=7517, Bruce says it is getting smoky but we should be about 100 m north of Lions Head.

2034
Top lab says Lions Head is 200 m behind us. We are too low on power to go back so we head on toward Raven.

2038
x=5342, y=7395, basalt rubble.

2040
doppler is hung up. Nav computer completely gone. Must rely on top lab LBL.

2043
Doppler reset to top lab LBL. X=5382, y=7269 but Doppler doesn’t track so top lab is our guide dog.
2053
x=5283, y=7095, mostly rubble, occasionally an intact pillow.
2055
x=5251, y=7949, basalt rubble

2058
x=5226, y=6990 still traversing along a ridge that drops off quickly to port. That fix is probably bogus, we are relying on top lab.

2103
Pat says we are 100 m away on hdg of 200.

2108
Have found an extinct sulfide on the edge of the ridge. Find four or five in this area which we believe is the Raven site.

2110
Still looking around Raven area for something active.

2116
I ask Bruce to collect a basalt sample but it turns out to be an extinct sulfide fro the raven area.
2118
weights away

